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/FD: Detector Assembly Progress )

ST Ve
@ The Intensity Frontier NOVA Far Detector Assembly Progress

Status Date: 29JUL14

.. Last Di-Block S
Instrumented
July 29, 2014

v" Last diblock instrumented Jul 29, 2014!
v" All diblocks cooled down on Aug 7, 2014!
v' First run with final configuration taken Aug 8, 2014!

\1 v" We found neutrino candidates in all superblocks! e
8/11/2014
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FarDet: Data-Taking
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Partition I (status Fri/Sat)

" diblocks 1 — 14 (over weekend)
" Running ~99% active,

" ALL COOLED, HYV at full gain

values.
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Partition 2 (current status)

m 3rd —ary oil top-up for DiB 13 side

" Water leak (Sat eve) recovery on DiB-01
" Repairs of non-cooled and noisy parts

on diblocks 12-14
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FD: DAQ Uptime
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> Last week we better one, 79% of uptime (4 weeks average 75%).
» On Mon we started swapped out old kind of APDs (aimed for NearDet) to new ones
> This with other repairs resulted to many re—configurations of the physics partitions and
interlock alarms with consequences of timing cooling issues through Tue and Wed.
» OnThu we cooled last part of detector and applied new HV settings for all repairs.
» On Sat the ‘bad’ water leak occurred on the side of DiB-12, kicked down 14-DB run.
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/FD: POTs delivered/recorded e
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» Beam was stable for a week , we used it with 78% efficiency.
> We recorded in average 0.97e18 (from 1.25) pots/ day with.
> After the issues we started 14-diblock run on Fri afternoon with very good

efficiency interrupted by the water leak on Sat late evening.
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NearDet: Detector Assembly Progress "
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NOVA Near Detector (DETID 1)
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BLUE = FEB only installed @ BROWN = APD installed

v 599/631 APD installed — 95% completed, DiB 02-04 and top of
diblock 01 installed, expect to finish APD installation this week!
v We are taking data on 1 partition: APDs installed, HV ON.

v’ Partition 2 with APDs is running as cold!
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/Summary i

AT A
FarDet:

e Far detector Completed and 14-diblock (cooled) running
started last week.

* Interrupted by the water leak on Sat.

e We have closed the book of the Far detector assembly and
are just opening the doors to retrofitting which should us
move over water leaks in future.

NearDet:
e APD installation about finish on DiB-01, total of 95%.

e We started to run all partitions in 24 /7 shifter mode.

¢ We run detector in cooled mode since last week.
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